%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% wagering.tex
% source file for ACM EC 2008 paper
% "Self-Financed Wagering Mechanisms for Forecasting"
% by Nicolas Lambert, John Langford, Jennifer Wortmann,
% Yiling Chen, Daniel Reeves, Yoav Shoham, David Pennock
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\documentclass{sig-alternate}

\usepackage{amsmath}
\usepackage{amsfonts}
\usepackage{amssymb}
\usepackage{dsfont}
\usepackage{tabularx}
\usepackage{booktabs}
\usepackage{bm}

\sloppy

\newcommand{\footnoteremember}[2]{\footnote{#2}\newcounter{#1}\setcounter{#1}{\value{footnote}}}
\newcommand{\footnoterecall}[1]{\footnotemark[\value{#1}]}

\DeclareMathOperator*{\argmin}{{arg\,min}}
\DeclareMathOperator*{\argmax}{{arg\,max}}
\newcommand{\one}{\mathds{1}}

% use ACM'own definition and theorem styles

\newtheorem{theorem}{Theorem}
\newtheorem{corollary}{Corollary}
\newtheorem{proposition}{Proposition}
\newtheorem{lemma}{Lemma}
\newdef{definition}{Definition}
\newdef{example}{Example}

% START of bold fonts
\newcommand{\bfA}{\mathbf{A}}
\newcommand{\bfB}{\mathbf{B}}
\newcommand{\bfC}{\mathbf{C}}
\newcommand{\bfD}{\mathbf{D}}
\newcommand{\bfE}{\mathbf{E}}
\newcommand{\bfF}{\mathbf{F}}
\newcommand{\bfG}{\mathbf{G}}
\newcommand{\bfH}{\mathbf{H}}
\newcommand{\bfI}{\mathbf{I}}
\newcommand{\bfJ}{\mathbf{J}}
\newcommand{\bfK}{\mathbf{K}}
\newcommand{\bfL}{\mathbf{L}}
\newcommand{\bfM}{\mathbf{M}}
\newcommand{\bfN}{\mathbf{N}}
\newcommand{\bfO}{\mathbf{O}}
\newcommand{\bfP}{\mathbf{P}}
\newcommand{\bfQ}{\mathbf{Q}}
\newcommand{\bfR}{\mathbf{R}}
\newcommand{\bfS}{\mathbf{S}}
\newcommand{\bfT}{\mathbf{T}}
\newcommand{\bfU}{\mathbf{U}}
\newcommand{\bfV}{\mathbf{V}}
\newcommand{\bfW}{\mathbf{W}}
\newcommand{\bfX}{\mathbf{X}}
\newcommand{\bfY}{\mathbf{Y}}
\newcommand{\bfZ}{\mathbf{Z}}
\newcommand{\bfa}{\mathbf{a}}
\newcommand{\bfb}{\mathbf{b}}
\newcommand{\bfc}{\mathbf{c}}
\newcommand{\bfd}{\mathbf{d}}
\newcommand{\bfe}{\mathbf{e}}
\newcommand{\bff}{\mathbf{f}}
\newcommand{\bfg}{\mathbf{g}}
\newcommand{\bfh}{\mathbf{h}}
\newcommand{\bfi}{\mathbf{i}}
\newcommand{\bfj}{\mathbf{j}}
\newcommand{\bfk}{\mathbf{k}}
\newcommand{\bfl}{\mathbf{l}}
\newcommand{\bfm}{\mathbf{m}}
\newcommand{\bfn}{\mathbf{n}}
\newcommand{\bfo}{\mathbf{o}}
\newcommand{\bfp}{\mathbf{p}}
\newcommand{\bfq}{\mathbf{q}}
\newcommand{\bfr}{\mathbf{r}}
\newcommand{\bfs}{\mathbf{s}}
\newcommand{\bft}{\mathbf{t}}
\newcommand{\bfu}{\mathbf{u}}
\newcommand{\bfv}{\mathbf{v}}
\newcommand{\bfw}{\mathbf{w}}
\newcommand{\bfx}{\mathbf{x}}
\newcommand{\bfy}{\mathbf{y}}
\newcommand{\bfz}{\mathbf{z}}
% END of bold fonts

% START of script fonts
\newcommand{\scA}{\mathcal{A}}
\newcommand{\scB}{\mathcal{B}}
\newcommand{\scC}{\mathcal{C}}
\newcommand{\scD}{\mathcal{D}}
\newcommand{\scE}{\mathcal{E}}
\newcommand{\scF}{\mathcal{F}}
\newcommand{\scG}{\mathcal{G}}
\newcommand{\scH}{\mathcal{H}}
\newcommand{\scI}{\mathcal{I}}
\newcommand{\scJ}{\mathcal{J}}
\newcommand{\scK}{\mathcal{K}}
\newcommand{\scL}{\mathcal{L}}
\newcommand{\scM}{\mathcal{M}}
\newcommand{\scN}{\mathcal{N}}
\newcommand{\scO}{\mathcal{O}}
\newcommand{\scP}{\mathcal{P}}
\newcommand{\scQ}{\mathcal{Q}}
\newcommand{\scR}{\mathcal{R}}
\newcommand{\scS}{\mathcal{S}}
\newcommand{\scT}{\mathcal{T}}
\newcommand{\scU}{\mathcal{U}}
\newcommand{\scV}{\mathcal{V}}
\newcommand{\scW}{\mathcal{W}}
\newcommand{\scX}{\mathcal{X}}
\newcommand{\scY}{\mathcal{Y}}
\newcommand{\scZ}{\mathcal{Z}}
\newcommand{\sca}{\mathcal{a}}
\newcommand{\scb}{\mathcal{b}}
\newcommand{\scc}{\mathcal{c}}
\newcommand{\scd}{\mathcal{d}}
\newcommand{\sce}{\mathcal{e}}
\newcommand{\scf}{\mathcal{f}}
\newcommand{\scg}{\mathcal{g}}
\newcommand{\sch}{\mathcal{h}}
\newcommand{\sci}{\mathcal{i}}
\newcommand{\scj}{\mathcal{j}}
\newcommand{\sck}{\mathcal{k}}
\newcommand{\scl}{\mathcal{l}}
\newcommand{\scm}{\mathcal{m}}
\newcommand{\scn}{\mathcal{n}}
\newcommand{\sco}{\mathcal{o}}
\newcommand{\scp}{\mathcal{p}}
\newcommand{\scq}{\mathcal{q}}
\def\scr\mathcal{r}
\newcommand{\scs}{\mathcal{s}}
\newcommand{\sct}{\mathcal{t}}
\newcommand{\scu}{\mathcal{u}}
\newcommand{\scv}{\mathcal{v}}
\newcommand{\scw}{\mathcal{w}}
\newcommand{\scx}{\mathcal{x}}
\newcommand{\scy}{\mathcal{y}}
\newcommand{\scz}{\mathcal{z}}
% END of script fonts

% START of blackboard fonts
\newcommand{\bbA}{\mathbb{A}}
\newcommand{\bbB}{\mathbb{B}}
\newcommand{\bbC}{\mathbb{C}}
\newcommand{\bbD}{\mathbb{D}}
\newcommand{\bbE}{\mathbb{E}}
\newcommand{\bbF}{\mathbb{F}}
\newcommand{\bbG}{\mathbb{G}}
\newcommand{\bbH}{\mathbb{H}}
\newcommand{\bbI}{\mathbb{I}}
\newcommand{\bbJ}{\mathbb{J}}
\newcommand{\bbK}{\mathbb{K}}
\newcommand{\bbL}{\mathbb{L}}
\newcommand{\bbM}{\mathbb{M}}
\newcommand{\bbN}{\mathbb{N}}
\newcommand{\bbO}{\mathbb{O}}
\newcommand{\bbP}{\mathbb{P}}
\newcommand{\bbQ}{\mathbb{Q}}
\newcommand{\bbR}{\mathbb{R}}
\newcommand{\bbS}{\mathbb{S}}
\newcommand{\bbT}{\mathbb{T}}
\newcommand{\bbU}{\mathbb{U}}
\newcommand{\bbV}{\mathbb{V}}
\newcommand{\bbW}{\mathbb{W}}
\newcommand{\bbX}{\mathbb{X}}
\newcommand{\bbY}{\mathbb{Y}}
\newcommand{\bbZ}{\mathbb{Z}}
\newcommand{\bba}{\mathbb{a}}
\newcommand{\bbb}{\mathbb{b}}
\newcommand{\bbc}{\mathbb{c}}
\newcommand{\bbd}{\mathbb{d}}
\newcommand{\bbe}{\mathbb{e}}
\newcommand{\bbf}{\mathbb{f}}
\newcommand{\bbg}{\mathbb{g}}
\newcommand{\bbh}{\mathbb{h}}
\newcommand{\bbi}{\mathbb{i}}
\newcommand{\bbj}{\mathbb{j}}
\newcommand{\bbk}{\mathbb{k}}
\newcommand{\bbl}{\mathbb{l}}
\newcommand{\bbm}{\mathbb{m}}
\newcommand{\bbn}{\mathbb{n}}
\newcommand{\bbo}{\mathbb{o}}
\newcommand{\bbp}{\mathbb{p}}
\newcommand{\bbq}{\mathbb{q}}
\newcommand{\bbr}{\mathbb{r}}
\newcommand{\bbs}{\mathbb{s}}
\newcommand{\bbt}{\mathbb{t}}
\newcommand{\bbu}{\mathbb{u}}
\newcommand{\bbv}{\mathbb{v}}
\newcommand{\bbw}{\mathbb{w}}
\newcommand{\bbx}{\mathbb{x}}
\newcommand{\bby}{\mathbb{y}}
\newcommand{\bbz}{\mathbb{z}}
% END of blackboard fonts

\title{Self-Financed Wagering Mechanisms for Forecasting\titlenote{Part of this work was done while N.~Lambert and J.~Wortman were at Yahoo! Research, New York.}}

\numberofauthors{6}
\author{
Nicolas Lambert\\
\affaddr{Stanford University}\\
\alignauthor John Langford\\
\affaddr{Yahoo!\ Research}\\
\alignauthor Jennifer Wortman \\
\affaddr{University of Pennsylvania}\\
\alignauthor Yiling Chen\\
\affaddr{Yahoo!\ Research}\\
\and
\alignauthor Daniel Reeves\\
\affaddr{Yahoo!\ Research}\\
\alignauthor Yoav Shoham \\
\affaddr{Stanford University}\\
\alignauthor David M.\ Pennock\\
\affaddr{Yahoo!\ Research}\\
}

\begin{document}
\begin{thebibliography}{10}

\bibitem{ali1977}
M.~Ali.
\newblock {Probability and Utility Estimates for Racetrack Bettors}.
\newblock {\em The Journal of Political Economy}, 85(4):803--815, 1977.

\bibitem{chen2007b}
Y.~Chen, L.~Fortnow, E.~Nikolova, and D.~Pennock.
\newblock {Betting on permutations}.
\newblock {\em Proceedings of the 8th ACM Conference on Electronic Commerce},
  pages 326--335, 2007.

\bibitem{chen2007}
Y.~Chen, D.~Reeves, D.~Pennock, R.~Hanson, L.~Fortnow, and R.~Gonen.
\newblock Bluffing and strategic reticence in prediction markets.
\newblock In {\em WINE}, pages 70--81, 2007.

\bibitem{economides2005}
N.~Economides and J.~Lange.
\newblock {A Parimutuel Market Microstructure for Contingent Claims Trading}.
\newblock {\em European Financial Management}, 11(1):25--49, 2005.

\bibitem{fortnow2005}
L.~Fortnow, J.~Kilian, D.~Pennock, and M.~Wellman.
\newblock {Betting Boolean-style: a framework for trading in securities based
  on logical formulas}.
\newblock {\em Decision Support Systems}, 39(1):87--104, 2005.

\bibitem{gneiting2007}
T.~Gneiting and A.~Raftery.
\newblock {Strictly proper scoring rules, prediction and estimation}.
\newblock {\em Journal of the American Statistical Association},
  102(477):359--378, 2007.

\bibitem{hanson2003}
R.~Hanson.
\newblock {Combinatorial Information Market Design}.
\newblock {\em Information Systems Frontiers}, 5(1):107--119, 2003.

\bibitem{johnstone2007}
D.~Johnstone.
\newblock {The Parimutuel Kelly Probability Scoring Rule}.
\newblock {\em Decision Analysis}, 4(2):66, 2007.

\bibitem{kehr2001}
C.-H. Kehr, J.~P. Krahnen, and E.~Theissen.
\newblock The anatomy of a call market.
\newblock {\em Journal of Financial Intermediation}, 10(3-4):249--270, 2001.

\bibitem{kilgour2004}
D.~Kilgour and Y.~Gerchak.
\newblock {Elicitation of Probabilities Using Competitive Scoring Rules}.
\newblock {\em Decision Analysis}, 1(2):108--113, 2004.

\bibitem{lambert2007}
N.~Lambert, D.~Pennock, and Y.~Shoham.
\newblock {Eliciting Properties of Probability Distributions}.
\newblock {\em Proceedings of the 9th ACM Conference on Electronic Commerce}, 2008.

\bibitem{milgrom1982}
P.~Milgrom and N.~Stokey.
\newblock Information, trade, and common knowledge.
\newblock {\em Journal of Economic Theory}, 1982.

\bibitem{peters2007}
M.~Peters, A.~Man-Cho~So, and Y.~Ye.
\newblock Pari-mutuel markets: Mechanisms and performance.
\newblock In {\em WINE}, pages 82--95, 2007.

\bibitem{savage1971}
L.~Savage.
\newblock {Elicitation of Personal Probabilities and Expectations}.
\newblock {\em Journal of the American Statistical Association},
  66(336):783--801, 1971.

\bibitem{thaler1988}
R.~H. Thaler and W.~T. Ziemba.
\newblock Anomalies: Parimutuel betting markets: Racetracks and lotteries.
\newblock {\em Journal of Economic Perspectives}, 2(2):161--174, 1988.

\bibitem{winkler1996}
R.~Winkler, J.~Mu{\~n}oz, J.~Cervera, J.~Bernardo, G.~Blattenberger, J.~Kadane,
  D.~Lindley, A.~Murphy, R.~Oliver, and D.~R{\'\i}os-Insua.
\newblock {Scoring rules and the evaluation of probabilities}.
\newblock {\em TEST}, 5(1):1--60, 1996.

\bibitem{wolfers2004}
J.~Wolfers and E.~Zitzewitz.
\newblock {Prediction Markets}.
\newblock {\em The Journal of Economic Perspectives}, 18(2):107--126, 2004.

\bibitem{wolfers2006}
J.~Wolfers and E.~Zitzewitz.
\newblock {Prediction Markets in Theory and Practice}.
\newblock {\em Working paper}, 2006.

\end{thebibliography}


\conferenceinfo{EC'08,} {July 8--12, 2008, Chicago, Illinois, USA.}
\CopyrightYear{2008}
\crdata{978-1-60558-169-9/08/07}

\maketitle

\begin{abstract}
We examine a class of wagering mechanisms designed to elicit truthful predictions from a group of people without requiring any outside subsidy. We propose a number of desirable properties for wagering mechanisms, identifying one mechanism---weighted-score wagering---that satisfies all of the properties. Moreover, we show that a single-parameter generalization of weighted-score wagering is the only mechanism that satisfies these properties. We explore some variants of the core mechanism based on practical considerations.
\end{abstract}

\category{J.4}{Social and Behavioral Sciences}{Economics}

\terms{Economics, Theory}

\keywords{Group forecasting, mechanism design, prediction markets}

\section{Introduction}

Consider a group of people with different estimates about an uncertain variable, for example different probability estimates or different quantile estimates. We seek mechanisms to induce all the group members to truthfully reveal their estimates to each other.

If some patron is willing to pay for the group's information, then the patron can use any number of well-known \emph{scoring rule} payment functions that give group members the incentive to truthfully report their estimates~\cite{winkler1996,savage1971}. However, if no patron is willing to subsidize the process, as we assume in this paper, then self-financed or budget-balanced mechanisms are needed.

Prediction markets~\cite{wolfers2004} can be run in a self-financed way, for example by employing double auction or parimutuel mechanisms. Such markets induce aggregate estimates~\cite{wolfers2006,hanson2003}, but do not induce truthful individual estimates~\cite{chen2007}. An informed trader may simply lack the wealth necessary to move the market to express a belief.  Similarly, if a trader has a belief that matches the current market price, there is no incentive to participate. Additionally, double auctions are prone to the thin-market problem: with few traders, the volume of transactions is insufficient attract other traders, hindering price discovery.

Kilgour and Gerchak~\cite{kilgour2004} also propose an adaptation of scoring rules for groups that are budget-balanced. However, they require that all group members wager the same amount of money. Recently, Johnstone~\cite{johnstone2007} adapted the Kilgour-Gerchak scoring rules to account for different wagers, but the resulting mechanism is not truthful. Newsfutures.com employs a form of competitive forecasting for continuous random variables that works well in practice, though is not truthful.

In this paper, we take an axiomatic approach, compiling a list of desirable properties in Section~\ref{sec:properties} that arguably any mechanism should have: budget balance, anonymity, truthfulness, normality, sybilproofness, individual rationality, and monotonicity. In Section~\ref{sec:preliminaries}, we construct a general space of mechanisms called \emph{wagering mechanisms} which contains many known approaches, including call markets and parimutuel markets. In Section~\ref{sec:weightedscore}, we identify the weighted-score mechanism, a generalization of Kilgour-Gerchak scoring rules, adapted to the elicitation of general distribution properties and with weights a function of amount wagered. We prove that weighted-score wagering satisfies all seven properties.

A natural question to ask is whether there are other mechanisms satisfying this set of properties.  We prove that the answer is \emph{no}. More precisely, in Section~\ref{sec:uniqueness} we prove that the mechanisms that satisfy the first five properties are exactly the weighted-score mechanisms, parameterized by the total amount wagered. Then, relaxing some of the properties, in Section~\ref{sec:discussion} we explore variants of the core mechanism based on practical considerations.


\section{Preliminaries}
 \label{sec:preliminaries}

\subsection{Model}
\label{subsec:model}

We consider a wagering setting involving a \emph{principal} and a finite set of agents, or \emph{players} $\scN = \{ 1, \dots, n \}$. The principal is interested in eliciting certain information on a given random experiment, for example a horse race or the election of a political candidate. We denote by $\Omega$ the set of possible outcomes. We assume that the outcome of the random experiment is drawn according to some distribution unknown to the principal, but on which players form beliefs. We model information of interest as \emph{distribution properties}, such as event probabilities or mean, defined in Section~\ref{subsec:dp}.

Players participate in a \emph{wagering mechanism}, formalized in the next subsection. Wagering mechanisms are essentially one-shot games in which players wager some money and make a prediction. After the realization of the experiment, each player receives a payout that is a function of her performance relative to the performance of other players.

We assume that players are risk neutral and seek to maximize their expected payout with respect to their belief. We argue that linear utility is a reasonable assumption when the money at stake remains small enough. We further assume that players choose their wager up front, before deciding on their prediction.\footnote{The amount wagered may depend on the player's belief/knowledge.  For example, the more confident the player is about some outcome, the more she may want to wager.} We believe that such a behavior is plausible in a context where players have fixed budgets. In particular, this is consistent with the model and findings of Ali~\cite{ali1977} in the case of horse-race parimutuel betting. Ali showed that amounts players are willing to bet are typically small and player utility varies approximately linearly in the range of amounts being bet.

\subsection{Wagering Mechanisms}
\label{subsec:wm}

Wagering mechanisms, which cover a large class of one-shot mechanisms, allow players to specify a wager as part of their actions. Wagering mechanisms operate in two steps. In the first step, each player announces a report chosen from a certain \emph{set of possible reports} $\scR$ and wagers any positive amount of money. Wagers are deposited in a common pot. The set of allowed reports $\scR$ may take various forms, for example, in the case of a horse race, players may be asked to specify a winning horse, or may be asked to lay out the probability of winning for each competing horse. In a second step, after the realization of the experiment, the common pot is divided among the players according to their performance, based on their reports and the true outcome of the experiment. When the mechanism is not budget-balanced, players may receive a bonus, or may have to pay a tax or fee. Using the vector notation $\bfx = (x_1, \dots, x_n)$, each player $i$ reporting $r_i$ and wagering $m_i$ gets a nonnegative payout $\Pi_i(\bfr, \bfm, \omega)$ specified by the mechanism. Payouts depend on reports and wagers, and on the true outcome of the experiment $\omega$. Player $i$'s net profit is thus $\Pi_i(\bfr, \bfm, \omega) - m_i$.

We assume that a player who wagers zero gets zero payout, so without loss of generality, we can use the set of natural numbers to represent the set of players $\scN = \bbN$ since nonparticipation is equivalent to a zero bid.\footnote{This notational change is useful to define wagering mechanisms with a varying number of players.} The formal definition of wagering mechanism is given below.

\begin{definition}
A \emph{Wagering Mechanism} is a tuple $(\scR, \Omega, \Pi)$ together with a set of players $\bbN$, where $\scR$ is the set of allowed reports, $\Omega$ is the outcome space, and $\Pi = (\Pi_i(\bfr, \bfm, \omega))_{i \in \bbN}$ is the vector of payout functions $\Pi_i: \scR^\bbN \times [0,+\infty)^\bbN \times \Omega \mapsto [0, +\infty)$, with $\Pi_i(\bfr, \bfm, \omega) = 0$ if $m_i = 0$.
\end{definition}

Wagering mechanisms include several well-known instances of betting mechanisms, in particular, parimutuel betting markets \cite{ali1977, thaler1988}. In a parimutuel market, players wager on mutually exclusive and exhaustive events $E_1, \dots, E_m$ (where, for example, $E_i = $ ``horse $i$ wins the race''). Players lose their wagers when the true outcome is not what they bet, while winning players share the total money wagered in proportion to their own wager. Such a market is a wagering mechanism with the set of reports $\scR = \{E_1, \dots, E_m\}$, and a payout
\begin{equation*}
\Pi_i = \one_{\omega \in r_i} \frac{m_i}{\sum_j m_j \one_{\omega \in
r_j}} \sum_j m_j ~,
\end{equation*}
when player $i$ reports $r_i $ and wagers $m_i$ dollars, and where $\one_{\omega \in A}$ is 1 if $\omega \in A$ and 0 otherwise.

Call markets \cite{kehr2001,peters2007,economides2005} can also be viewed as wagering mechanisms. In a typical call market for binary events, participants trade on contingent contracts, each contract corresponding to an outcome, paying off \$1 if the outcome becomes true and \$0 otherwise. Traders may submit orders, which indicate the maximum price they are willing to pay for a certain contract and a maximum amount of money to spend. Whenever possible, the market matches orders for one outcome with orders for the opposite one. When the space of outcomes is large, more exotic betting languages can be used to allow, for example, betting on combinations of events. This includes Boolean betting where players bet on Boolean formulas of events \cite{fortnow2005} and permutation betting where players bet on properties of the final ranking of competing candidates \cite{chen2007b}. Such combinatorial betting may use call markets with multilateral order matching to clear bets, which can be modeled as wagering mechanisms. However the payout functions are often very complicated. Furthermore, these instances of wagering mechanisms do not admit a dominant strategy; rather, players make reports conditionally on their beliefs on other players' reports. At best, when the range of possible reports is limited to outcomes or events, the mechanism may allow a partial specification of individual subjective probabilities such as: the (subjective) probability is greater than some threshold. However this is an incomplete specification and less natural than simply stating a belief.

\subsection{Distribution Properties}
\label{subsec:dp}

Distribution properties, introduced in Lambert et al.~\cite{lambert2007}, are a convenient way to model information on probability distributions. A \emph{distribution property} $\Gamma(P)$ is defined as a function that assigns a real value to any probability distribution $P$ in a given convex domain. By assumption, the domain must contain the true distribution of the random experiment under consideration. For example, in the case of a continuous outcome, the domain of the median would be the set of continuous densities with full support, the domain of the expectation the set of distributions with finite first moment.

Common distribution properties include the probability of an event, the expectation, the variance, medians/quantiles, moments, indicators of the symmetry of the distribution (skewness), and dispersion (kurtosis). For instance, the property $\Gamma$ corresponding to the probability of an event $A$ is $\Gamma(P) =P(A)$, that of the median of a random variable $X$ is $\Gamma(P) = \sup_m \{ m : P(X<m) < 1/2\}$. As in Lambert et al.~\cite{lambert2007}, we say that a report $r$ (respectively, a probability $P$) is admissible when $r$ (respectively $\Gamma(P)$) falls in the interior of $\Gamma$'s range, which is an open interval for continuous properties. We are interested in wagering mechanisms wherein players report sets or vectors of property values. For example, a single probability of a given event, the full distribution for a finite set of outcomes, or a pair \emph{(expectation, variance)}.

Similar to scoring rules~\cite{winkler1996}, Lambert et al.~\cite{lambert2007} define reward functions that truthfully elicit single properties or sets of properties from a risk neutral agent as follows.
\begin{definition}
A score function for a vector of distribution properties $\Gamma = (\Gamma_1, \dots, \Gamma_k)$ is a real-valued function $s(r, \omega)$, with $r=(r_1, \dots, r_k)$ and $r_i$ the report for property $\Gamma_i$, and $\omega$ the outcome. It is called \emph{strictly proper} when
\begin{equation}\label{eq:strictproper}
E_P[s(r, \omega)] < E_P[s(\Gamma(P), \omega)]
\end{equation}
for all admissible probability $P$ and vector $r = (r_1, \dots, r_k)
\neq (\Gamma_1(P), \dots, \Gamma_k(P))$.
\end{definition}
Here and throughout the remainder of the paper, $E_P[X]$ denotes the expectation of $X$ when the outcome $\omega$ is distributed according to the distribution $P$. For convenience, we identify single properties as vectors of properties with a single element ($k=1$).

\section{Weighted-score mechanisms}
\label{sec:weightedscore}

In this section we present a specific subclass of wagering mechanisms called \emph{weighted-score mechanisms}. These mechanisms are weighted mixtures of strictly proper score functions. Rewards are determined by (1) the relative performance of the players, as in the scoring rules of Kilgour and Gerchak~\cite{kilgour2004}, and (2) the amounts wagered, as in parimutuel betting markets.

\begin{definition}
A \emph{weighted-score mechanism} is a wagering mechanism $(\Omega, \scR, \Pi)$ associated with a vector of properties $\Gamma = (\Gamma_1, \dots, \Gamma_k)$. Here $\Omega$ is the space of possible outcomes, $\scR = (a_1, b_1) \times \cdots \times (a_k, b_k)$ where $(a_i, b_i)$ is an interval of possible values for $\Gamma_i$, and $\Pi$ is the vector of payout functions with the payout to bettor $i$ defined as
\begin{equation*}
\Pi_i(\bfr, \bfm, \omega) = m_i\left(1+
s(r_i, \omega) - \frac{\sum_j s(r_j, \omega) m_j}{\sum_j m_j}\right)
\end{equation*}
where $s$ is a strictly proper score function for $\Gamma$ taking values in the interval $[0,1]$.
\end{definition}

We need to choose $s$ so as verify Equation~(\ref{eq:strictproper}). Fortunately, there already exists a wealth of such functions for common properties~\cite{winkler1996}. Savage~\cite{savage1971} gives a simple characterization of strictly proper score functions for probabilities and expectations of random variables, and examples of strictly proper scores for quantiles appear in Gneiting and Raftery~\cite{gneiting2007}. For example, to elicit the probability of an event $A$, one may use the quadratic score $s(p,\omega) = 1-(\one_{\omega \in A}-p)^2$. To elicit the median of a continuous variable in the interval $[\alpha, \beta]$, one can use $s(m,\omega) = 1-|m-\omega|/(\beta-\alpha)$. Table~\ref{tbl:examples} gives examples of weighted-score mechanisms to elicit the probability of a binary event, and the expectation and median of a continuous random variable.

Weighted-score mechanisms are interesting and valuable because they satisfy many desirable properties. These are discussed in detail in the next section. In Section~\ref{sec:uniqueness}, we will see that the core properties are verified by the slightly larger class of weighted-score mechanisms parameterized by the total money wagered, and that those are the \emph{only} mechanisms satisfying these properties.

Note that when one elicits probabilities of a binary event (when $\Gamma(P)=P(A)$ for some event $A$), and when all wagers are equal, payouts are proportional to those given by the KG-scoring rules~\cite{kilgour2004}. As we shall see in Section~\ref{sec:uniqueness}, the multiplicative factor, $(n-1)/n$ when there are $n$ participants, is necessary to induce sybilproofness.

Weighted-score mechanisms can be used for \emph{free} forecasting. They reveal truthful predictions of distribution properties from a group of experts, without requiring any outside subsidy. Betting is another natural field of application. Indeed, common betting mechanisms are limited in several ways. For example, parimutuel markets do not work properly when all bettors agree that one event is much more likely than its alternatives. These markets are also not applicable with continuous outcomes, such as date and time. Besides the limited range of bets allowed prevents the full exploitation of one's information. Weighted-score mechanisms overcome these issues.

\begin{table*}
\centering
\caption{Some examples of weighted-score mechanisms.}\label{tbl:examples}
\begin{tabular}{p{0.18\textwidth}p{0.3\textwidth}@{\hspace{0.03\textwidth}}p{0.4\textwidth}}
\toprule
\textbf{Space of outcomes:}
&
\textbf{Information elicited:}
&
\textbf{Payout for player $\bm{i}$:}
\\
\midrule
\footnotesize Any space $\Omega$
&
\footnotesize Probability of an event $A \subset \Omega$
&
\footnotesize
$\begin{displaystyle}
m_i + m_i \frac{\sum_j m_j (\one_{\omega \in A} - r_j)^2}{\sum_j m_j} - m_i (\one_{\omega \in A} - r_i)^2
\end{displaystyle}$
\\
\footnotesize $\Omega = [\alpha, \beta]$
&
\footnotesize Expectation of the outcome
&
$\begin{displaystyle}
m_i + m_i \frac{\sum_j m_j (\omega - r_j)^2}{(\alpha-\beta)^2 \sum_j m_j} - m_i \frac{(\omega - r_i)^2}{(\alpha-\beta)^2}
\end{displaystyle}$
\\
\footnotesize $\Omega = [\alpha, \beta]$
&
\footnotesize Median of the outcome
&
\footnotesize
$\begin{displaystyle}
m_i + m_i \frac{\sum_j m_j |\omega - r_j|}{|\alpha-\beta|\sum_j m_j} - m_i \frac{|\omega - r_i|}{|\alpha-\beta|}
\end{displaystyle}$
\\
\bottomrule
\end{tabular}
\end{table*}

\section{Mechanism Properties}
\label{sec:properties}

\subsection{Desirable Properties}

We describe seven desirable properties for wagering mechanisms. Denote by $M = (\scR, \Omega, \Pi)$ a wagering mechanism, and let $\Gamma$ be a vector of properties $\Gamma=(\Gamma_1, \dots, \Gamma_k)$.

The first three properties are adapted from the axioms proposed by Kilgour and Gerchak \cite{kilgour2004}.

\renewcommand{\descriptionlabel}[1]{\hspace{\labelsep}\textsc{#1}}
\begin{description}
\item[Budget-Balance] $M$ is \emph{budget-balanced} if the market generates neither profit nor loss, i.e,
\begin{equation*}
\sum_i \Pi_i(\bfr, \bfm, \omega) = \sum_i m_i,
\end{equation*}
 for any vector of reports $\bfr$, any vector of money wagered $\bfm$, and any outcome $\omega$.

\item[Anonymity] $M$ is \emph{anonymous} if the payouts do not depend on the identity of the players. For any permutation $\sigma$ of $\bbN$, any player $i$, and any outcome $\omega$,
\begin{align*}
\Pi_i((r_i)_{i \in \bbN}, & (m_i)_{i \in \bbN}, \omega) = \\
& \Pi_{\sigma(i)}((r_{\sigma^{-1}(i)})_{i \in \bbN}, (m_{\sigma^{-1}(i)})_{i \in \bbN}, \omega) ~.
\end{align*}

\item[Truthfulness] $M$ is \emph{truthful} for $\Gamma$ when players maximize their expected payout when reporting true property values. For any player $i$, any admissible probability $P$, any set of reports of others $\bfr_{-i}$ and any vector of wagers $\bfm$,
\[
E_{P}[\Pi_i((\bfr_{-i}, \Gamma(P)), \bfm, \omega)] >
E_{P}[\Pi_i((\bfr_{-i}, r_i), \bfm, \omega)]
\]
is satisfied for all $r_i \neq \Gamma(P)$.
\end{description}

When a player's expected payout depends both on her own report and on reports of others, it is natural to interpret the player's payment as her performance relative to other players. The relative performance of a player should increase either when the player's absolute performance increases, or when the absolute performance of another player decreases. This property is captured by \emph{normality}.\footnote{Note that individual rationality, truthfulness and budget-balance do not necessarily imply normality for more than two players. For example, the mechanism of Section~\ref{subsec:adaptive} verifies all three properties but is not normal.}
\begin{description}
\item[Normality] A mechanism $M$ is \emph{normal} if, for any admissible probability distribution $P$, if any player $i$ changes her report, the changes of expected payouts $\Delta_j$, with respect to $P$, of any other player $j$ is null or of the opposite sign of the changes of expected payouts $\Delta_i$ of player $i$.
\end{description}
In electronic platforms, it is important to prevent players from manipulating identities. To reflect this requirement, we complement truthfulness by \emph{sybilproofness}, which ensures that payouts remain unchanged as a subset of players with the same reports manipulate user accounts, either by merging accounts or creating fake identities, or by doing transfers of wagers between them.
\begin{description}
\item[Sybilproofness] $M$ is \emph{sybilproof} if for any subset of players $\scS \subset \bbN$, for any reports $\bfr$ with $r_i = r_j$ for $i, j \in \scS$, for any vectors of wagers $\bfm, \bfm'$ such that $m_i = m'_i$ for $i \not\in \scS$ and $\sum_{i \in \scS} m_i = \sum_{i \in \scS} m'_i$, the following two conditions hold.

For all $i \not\in \scS$, for all $\omega$,
\[
\Pi_i(\bfr, \bfm, \omega) = \Pi_i(\bfr, \bfm', \omega) ~,
\]
and for all $\omega$,
\[
\sum_{i \in \scS} \Pi_i(\bfr, \bfm, \omega)
= \sum_{i \in \scS} \Pi_i(\bfr, \bfm', \omega) ~.
\]
\end{description}
The next property deals with the participation constraint. \emph{Individual rationality} ensures that players get a nonnegative expected profit with respect to their belief.
\begin{description}
\item[Individual Rationality] $M$ is \emph{individually rational} if for any player $i$, for any admissible probability distribution $P$, for any wager $m_i > 0$, there exists a report $r^*_i$ such that for any wagers $\bfm_{-i}$, and reports $\bfr_{-i}$ of other players, the expected profit of $i$ is nonnegative:
\begin{equation*}
E_{P}[\Pi_i((\bfr_{-i}, r^*_i), \bfm, \omega)-m_i] \geq 0
\end{equation*}
\end{description}
Milgrom and Stokey~\cite{milgrom1982} proved the \emph{no-trade theorem} which states that, under certain assumptions, agents should not trade with each other in a zero-sum market. The no-trade theorem requires that agents have the same prior, that they be Bayesian rational (performing Bayesian belief updates), and that their Bayesian rationality be common knowledge. Individual rationality as defined above relies on the assumption that players do not satisfy all three conditions required by the no-trade theorem, thus can serve as a participation constraint. Indeed, in practice, people do trade in zero-sum and even negative-sum markets, for example because they don't share the same priors, because they are over-confident or bounded rational, or because there are noisy traders in the market.

Our final property deals with the connection between wagers and profits. To increase participation incentives and larger wagers, it is preferable that a player making a positive expected profit under her own belief makes an even higher profit by increasing her wager. Similarly, a player who loses money on expectation should lose even more by raising her participation level. We call it \emph{monotonicity}.
\begin{description}
\item[Monotonicity] $M$ is \emph{monotonic} if for any player $i$, any admissible probability distribution $P$, any vector of reports $\bfr$, any vector of wagers $\bfm$, and any $M_i > m_i$, either
\begin{eqnarray*}
0 &<& E_{P}[\Pi_i(\bfr, (\bfm_{-i}, m_i), \omega)-m_i] \\
&<& E_{P}[\Pi_i(\bfr, (\bfm_{-i}, M_i), \omega)-M_i]
\end{eqnarray*}
or
\begin{eqnarray*}
0 &>& E_{P}[\Pi_i(\bfr, (\bfm_{-i}, m_i), \omega)-m_i] \\
&>& E_{P}[\Pi_i(\bfr, (\bfm_{-i}, M_i), \omega)-M_i] ~.
\end{eqnarray*}
\end{description}

\subsection{Attributes of weighted-score mechanisms}

We show that weighted-score mechanisms satisfy all the desired properties described above.
\begin{theorem}\label{th:wsattributes}
All weighted-score mechanisms are:
\begin{enumerate}\vspace{-4pt}
\item Budget-balanced, \vspace{-6pt}
\item Anonymous,\vspace{-6pt}
\item Truthful,\vspace{-6pt}
\item Sybilproof,\vspace{-6pt}
\item Normal,\vspace{-6pt}
\item Individually rational,\vspace{-6pt}
\item and Monotonic.
\end{enumerate}
\end{theorem}
Besides, due to their linear form, weighted-score mechanisms are also group strategyproof, in the sense that they incentivize honest behavior not only at the individual level but also at the group level: a group of players maximizes its expected global payout only when each of its members reports true property values.

\begin{proof}
We prove each property separately as follows..
\vspace{2pt}

\noindent\textbf{(1) Budget-Balance}
\vspace{2pt}
For any $\bfr$, $\bfm$, and $\omega$,
{\small
\begin{eqnarray*}
\sum_i \Pi_i(\bfr, \bfm, \omega) &=&  \sum_i m_i + \left( \sum_i s(r_i, \omega) m_i \right) \\
& & - \left( \sum_i m_i \right) \left( \frac{\sum_j s(r_j, \omega ) m_j}{\sum_j m_j} \right) \\
&=& \sum_i m_i ~.
\end{eqnarray*}
}
\vspace{2pt}

\noindent\textbf{(2) Anonymity}
\vspace{2pt}
Let $\sigma$ be any permutation of $\bbN$.  For any $\bfr$, $\bfm$, $\omega$, and $i$,
{\small
\begin{eqnarray*}
\lefteqn{
\Pi_{\sigma(i)}((r_{\sigma^{-1}(j)})_{j \in \bbN}, (m_{\sigma^{-1}(j)})_{j \in \bbN}, \omega)
}
\\
&=& m_{\sigma^{-1}(\sigma(i))} \left(1 + s(r_{\sigma^{-1}(\sigma(i))}, \omega)\right.
\\
& & \qquad \qquad \qquad\left. - \frac{\sum_j s(r_{\sigma^{-1}(j)}, \omega) m_{\sigma^{-1}(j)}}{\sum_j m_{\sigma^{-1}(j)}} \right)
\\
&=& m_i \left(1 + s(r_i, \omega) - \frac{\sum_j s(r_j, \omega) m_j}{\sum_j m_j} \right)
\\
&=& \Pi_i((r_j)_{j \in \bbN}, (m_j)_{j \in \bbN}, \omega) ~.
\end{eqnarray*}
}
\vspace{2pt}

\noindent\textbf{(3) Truthfulness}
\vspace{2pt}
For any $\bfr$, $\bfm$, $\omega$, $i$, and $P$,
{\small
\begin{eqnarray*}
E_P[\Pi_i(\bfr, \bfm, \omega)] &=& m_i \left( 1 + E_P[s(r_i, \omega)] \left(1 - \frac{m_i}{\sum_j m_j} \right) \right.
\\
& & \qquad \qquad \quad \left. - \frac{\sum_{j \neq i} E_P[s(r_j, \omega)] m_j}{\sum_j m_j} \right).
\end{eqnarray*}
}
Since $s$ is strictly proper for $\Gamma$, $E_P[s(r_i, \omega)]$ is maximized only at $r_i = \Gamma(P)$, so
{\small
\[
E_P[\Pi_i((\bfr_{-i}, r_i), \bfm, \omega)] < E_P[\Pi_i((\bfr_{-i}, \Gamma(P)), \bfm, \omega)]
\]
}
for all $r_i \neq \Gamma(P)$.
\vspace{2pt}

\noindent\textbf{(4) Sybilproofness}
\vspace{2pt}
Let $r$ be the common report of all $i \in \scS$. For any $i \not\in\scS$,
{\small
\begin{eqnarray*}
\lefteqn{
\Pi_i(\bfr, \bfm, \omega)
}
\\
&=& m_i \left(1 + s(r_i,\omega) \right.
\\
& & \quad - \frac{\sum_{j \not\in \scS} s(r_j,\omega) m_j + s(r,\omega) \sum_{j \in \scS} m_j}{\sum_{j \not\in \scS} m_j + \sum_{j \in \scS} m_j}
\\
&=& m'_i \left(1 + s(r_i,\omega)  \right.
\\
& & \quad
- \frac{\sum_{j \not\in \scS} s(r_j,\omega) m'_j + s(r,\omega) \sum_{j \in \scS} m'_j}{\sum_{j \not\in \scS} m'_j + \sum_{j \in \scS} m'_j}
\\
&=& \Pi_i(\bfr, \bfm', \omega).
\end{eqnarray*}
}
Additionally,
{\small
\begin{eqnarray*}
\lefteqn{
\sum_{i \in \scS} \Pi_i(\bfr, \bfm, \omega)
}
\\
&=& \sum_{i \in \scS} m_i \left(1 + s(r,\omega) \right.
\\
& & \quad
- \frac{\sum_{j \not\in \scS} s(r_j,\omega) m_j + s(r,\omega) \sum_{j \in \scS} m_j}{\sum_{j \not\in \scS} m_j + \sum_{j \in \scS} m_j}
\\
&=& \left(\sum_{i \in \scS} m'_i\right) \left(1 + s(r,\omega)  \right.
\\
& & \quad
- \frac{\sum_{j \not\in \scS} s(r_j,\omega) m'_j + s(r,\omega) \sum_{j \in \scS} m'_j}{\sum_{j \not\in \scS} m'_j + \sum_{j \in \scS} m'_j}
\\
&=& \sum_{i \in \scS} \Pi_i(\bfr, \bfm', \omega).
\end{eqnarray*}
}
\vspace{2pt}

\noindent\textbf{(5) Normality}
\vspace{2pt}
Let $\tilde{r}_i$ and $\tilde{\bfr} = (\bfr_{-i},\tilde{r}_i)$ be defined in such a way that
{\small
\[
E_P[\Pi_i(\tilde{\bfr}, \bfm,\omega)] > E_P[\Pi_i(\bfr, \bfm,\omega)].
\]
}
Then
{\small
\[
E_P[s(\tilde{r}_i, \omega)] > E_P[s(r_i, \omega)],
\]
}
and for $j \neq i$,
{\small
\begin{eqnarray*}
\lefteqn{
E_P[\Pi_j(\tilde{\bfr}, \bfm, \omega)] - E_P[\Pi_j(\bfr, \bfm,\omega)] } \\
&=& - \frac{m_i}{\sum_j m_j} (E_P[s(\tilde{r}_i, \omega)] - E_P[s(r_i, \omega)]) < 0 .
\end{eqnarray*}
}
Similarly, if $E_P[\Pi_i(\tilde{\bfr}, \bfm,\omega)] < E_P[\Pi_i(\bfr, \bfm,\omega)]$ then $E_P[\Pi_j(\tilde{\bfr}, \bfm, \omega)] - E_P[\Pi_j(\bfr, \bfm,\omega)] > 0$. This proves normality.
\vspace{2pt}

\noindent\textbf{(6) Individual rationality}
\vspace{2pt}
For fixed wagers $\bfm$ and a probability distribution $P$, $E_P[s(r_i, \omega)]$ is maximized when $r_i = \Gamma(P)$.  Thus, $E_P[s(r_j, \omega)] \leq E_P[s(\Gamma(P), \omega)]$ for all $j$.  We have
{\small
\[
\frac{\sum_j E_P[s(r_j, \omega)] m_j}{\sum_j m_j} \leq
E_P[s(\Gamma(P), \omega)].
\]
}
Hence,
{\small
\begin{align*}
E_P[\Pi_i(\bfr, \bfm, \omega)-m_i] & = m_i E_P[s(r_i, \omega)] -m_i
\frac{\sum_j E_P[s(r_j, \omega)] m_j}{\sum_j m_j} \\
 & \geq m_i [E_P s(r_i, \omega)] - m_i E_P[s(\Gamma(P), \omega)] \\
& = 0
\end{align*}
}
when player $i$ reports $r_i = \Gamma(P)$.
\vspace{2pt}

\noindent\textbf{(7) Monotonicity}
\vspace{2pt}
Let $\tilde{s}_i = E_P[s(r_i, \omega)]$.  Then
{\small
\[
E_P[\Pi_i(\bfr, \bfm, \omega)-m_i] =
m_i\left(\tilde{s}_i - \frac{\sum_j \tilde{s}_j m_j}{\sum_j
m_j}\right)
= \frac{a m_i}{m_i + b},
\]
}
where $a = \tilde{s}_i \sum_{j \neq i} m_j  - \sum_{j \neq i} \tilde{s}_j m_j$ and $b = \sum_{j \neq i} m_j$.

Because the value of $ax/(x+b)$ is positive and increases with $x$ when $a > 0$ and is negative and decreases with $x$ when $a < 0$, monotonicity holds.
\end{proof}

\section{Uniqueness of Weighted Score Mechanisms}
\label{sec:uniqueness}

Theorem~\ref{th:wsattributes} shows that the family of weighted score mechanisms satisfies a number of useful properties. In this section, we will show that weighted score mechanisms are unique in this sense. More precisely, weighted score mechanisms, parameterized by the total money wagered in the common pool, are the \emph{only} wagering mechanisms that are simultaneously budget-balanced, anonymous, truthful, normal, and sybilproof.

We start by characterizing the set of all truthful and normal wagering mechanisms, and progressively add the constraints of anonymity, budget-balance, and sybilproofness. This incidentally proves that the mechanisms introduced by Kilgour and Gerchak~\cite{kilgour2004} without subsidy are the only \emph{Competitive Prediction Schemes} to be truthful, normal, anonymous, and budget-balanced.

In the analysis that follows, we assume that $\Omega$ is finite, and consider elicitation of a single property $\Gamma$ (for example, the probability of a binary event). We assume $\Gamma$ is continuous and not locally constant,\footnote{With respect to the topology considered in Lambert et al.~\cite{lambert2007}.} and denote by $(a,b)$ the interval of admissible reports. As in Lambert et al.~\cite{lambert2007}, we define a distribution property to be elicitable when there exists a strictly proper score function for that property. We say that a function is smooth when it is twice continuously differentiable. In the sequel of this section, we consider mechanisms with smooth payouts, and when we refer to the term ``wagering mechanism'' we always mean ``wagering mechanism with smooth payouts''.

\subsection{Characterizing truthful and normal mechanisms}

The first characterization lemma shows that any wagering mechanism is truthful and normal when it is additively separable into strictly proper score functions.

\begin{lemma}\label{lem:tn}
A wagering mechanism is \emph{truthful} for $\Gamma$ and \emph{normal} if and only if its payouts are nonnegative and additively separable in the form
\begin{equation*}
\Pi_i(\bfr, \bfm, \omega) = m + f_{i,i}(r_i, \bfm, \omega) - \sum_{j \neq i} f_{i,j}(r_j, \bfm, \omega)
\end{equation*}
where for all $i$ and $j$, for any fixed value of $\bfm$, $f_{i,j}$ is a smooth strictly proper score function for $\Gamma$.
\end{lemma}

The proof will make use of the following lemma, whose proof is omitted due to lack of space.
\begin{lemma}\label{lem:decomposition}
If $f:(a,b)^n \mapsto \bbR$ is twice continuously differentiable, and if
%$\partial^2 f(x_1, \dots, x_n)/\partial x_i \partial x_j = 0$
\begin{equation*}
\frac{\partial^2 f(x_1, \dots, x_n)}{\partial x_i \partial x_j} = 0
\end{equation*}
for all $\bfx \in \bbR^n$ and all $i \neq j$, then there exists $f_i : (a,b) \mapsto \bbR$ such that
\begin{equation*}
f(x_1, \dots, x_n) = \sum_{i=1}^n f_i(x_i) ~.
\end{equation*}
\end{lemma}

\begin{proof}[(Lemma~\ref{lem:tn})]
The truthfulness and normality of any rule of this form follow from the linearity of expectation and $f_{i,j}$ being strictly proper scores for $\Gamma$. It remains to show that truthful and normal rule must be of this form.


Let $n$ be the number of participating players. Since we assume the wagers $m_1,\cdots,m_n$ are fixed, we can denote the payout function for $i$ as $\Pi_i(r_1, \dots, r_n, \omega) = \Pi_i(\bfr, \omega)$.

Let $P_i$ be the beliefs of bettor $i$.  By truthfulness,
\begin{equation}\label{eq:tn-1}
\Gamma(P_i) \in \argmax E_{P_i}[\Pi_i(\bfr_{-i}, \cdot, \omega)]
\end{equation}
for all $\bfr_{-i}$.  By normality, for any $k \neq i$,
\begin{equation}\label{eq:tn-2}
\Gamma(P_i) \in \argmin E_{P_i}[\Pi_j(\bfr_{-i}, \cdot, \omega)] ~.
\end{equation}

The first order condition in~(\ref{eq:tn-1}) and~(\ref{eq:tn-2}) gives that for all admissible beliefs $P_i$, for all $k \in \{1,\cdots,n\}$,
\[
\left. \frac{\partial E_{P_i}[\Pi_k(\bfr, \omega)]}{\partial r_i}
\right|_{r_i = \Gamma(P_i)} = 0 ~.
\]

Applying the same argument to any $j \neq i$ with beliefs $P_j$ gives us
\[
\left. \frac{\partial E_{P_j}[\Pi_k(\bfr, \omega)]}{\partial r_j}
\right|_{r_j = \Gamma(P_j)} = 0 ~.
\]

We now differentiate the left side of the these equations by $r_j$ and $r_i$ respectively to get
\[
\left. \frac{\partial^2 E_{P_i}[\Pi_k(\bfr, \omega)]}{\partial r_i
\partial r_j} \right|_{r_i = \Gamma(P_i)}
=
\left. \frac{\partial^2 E_{P_j}[\Pi_k(\bfr, \omega)]}{\partial r_j
\partial r_i} \right|_{r_j = \Gamma(P_j)} = 0 ~.
\]

Now consider admissible $P_i, P_j$ with different property values; in other words, $\Gamma(P_i) \neq \Gamma(P_j)$. Since $\Pi_k$ is smooth,
\[
\frac{\partial^2 \Pi_k}{\partial r_i \partial r_j} = \frac{\partial^2 \Pi_k}{\partial r_j \partial r_i} ~,
\]
so by linearity of expectation, for all admissible $P_i$ and $P_j$,
\[
E_{P_i}\left[\left. \frac{\partial^2 \Pi_k(\bfr, \omega)}{\partial r_i \partial r_j}\right|_{r_i = \Gamma(P_i), r_j = \Gamma(P_j)} \right] =
0 ~,
\]
and
\[
E_{P_j}\left[\left. \frac{\partial^2 \Pi_k(\bfr, \omega)}{\partial r_i \partial r_j}\right|_{r_i = \Gamma(P_i), r_j = \Gamma(P_j)} \right] =
0 ~.
\]

Let $r_i^* \neq r_j^*$ be admissible reports, and let $X$ be the random variable defined by
\begin{equation*}
X(\omega) = \left. \frac{\partial^2 \Pi_k(\bfr, \omega)}{\partial r_i \partial r_j} \right|_{r_i = r_i^*, r_j = r_j^*} ~.
\end{equation*}
The previous equations says that for all probabilities $P_i, P_j$ such that $\Gamma(P_i) = r_i^*$ and $\Gamma(P_j) = r_j^*$, we have $E_{P_i}[X] = E_{P_j}[X] = 0$.

Now, Theorem 2 of Lambert et al.~\cite{lambert2007} shows that elicitable properties may be expressed as linear constraints.  In particular, it shows that there exist random variables $Y_i$ and $Y_j$ such that
\begin{equation*}
E_P[Y_i] = 0 \Leftrightarrow \Gamma(P) = r_i^*
\end{equation*}
and
\begin{equation*}
E_P[Y_j] = 0 \Leftrightarrow \Gamma(P) = r_j^* ~.
\end{equation*}
Since $E_P[Y_i] = 0$ implies $\Gamma(P) = r_i^*$ which in turn implies that $E_P[X] = 0$, it must be the case that $X$ is proportional to $Y_i$.  By a similar argument, $X$ must be proportional to $Y_j$. If $X$ is not null, $Y_i$ and $Y_j$ are proportional, implying that $r_i^* = r_j^*$, which is a contradiction. Therefore it must be the case that $X(\omega)=0$ for all $\omega$, and for all admissible reports $r^*_i \neq r^*_j$,
\begin{equation*}
\left.\frac{\partial^2 \Pi_k(r_1, \dots, r_n, \omega)}{\partial r_i \partial r_j}
\right|_{r_i = r^*_i, r_j = r^*_j} = 0~.
\end{equation*}
This remains true when $r^*_i = r^*_j$ by continuity.

By Lemma~\ref{lem:decomposition}, this implies that there are functions $f_{k,i}$ such that
\[
\Pi_k(\bfr, \omega) = f_{k,k} (r_k, \omega) - \sum_{i \neq k}
f_{k,i}(r_i, \omega) ~.
\]

By truthfulness, $f_{k,k}$ must be a strictly proper score function for $\Gamma$; by normality, for every $i \neq k$, $f_{k, i}$ must be a strictly proper score function for $\Gamma$.
\end{proof}

\subsection{Adding Anonymity}

With anonymity added, we provide a necessary condition for the special case of identical wagers.

\begin{lemma}\label{lem:tna}
If a wagering mechanism is \emph{truthful} for $\Gamma$, \emph{normal}, and \emph{anonymous}, then there exist smooth functions $f$ and $g$ such that if every player wagers the same amount $m$, then for all $i$ the payout to $i$ is
\begin{equation*}
\Pi_i(\bfr, \bfm, \omega)
= m + f(r_i, m, M, \omega) - \sum_{j \neq i} g(r_j, m, M, \omega) ~,
\end{equation*}
where $M$ is the total amount wagered.  Furthermore, for any fixed values of $m$ and $M$, $f$ and $g$ must be strictly proper score functions for $\Gamma$.
\end{lemma}
The proof, omitted due to space restrictions, is based on successive use of the anonymity property.

\subsection{Adding budget-balance}

We now add budget-balance, and easily show the following, still considering the special case of identical wagers.

\begin{lemma}\label{lem:tnabb}
If a wagering mechanism is \emph{truthful} for $\Gamma$, \emph{normal}, \emph{anonymous}, and \emph{budget-balanced}, then there exists a smooth function $f$ such that if every agent wagers the same amount $m$, then payouts are
\begin{equation*}
\Pi_i(\bfr, \bfm, \omega) = m + f(r_i, m, M, \omega) -
\frac{1}{n-1}\sum_{j \neq i} f(r_j, m, M, \omega) ,
\end{equation*}
where $M$ is the total amount wagered.  Furthermore, for any fixed values of $m$ and $M$, $f$ must be a strictly proper score function for $\Gamma$.
\end{lemma}
The proof can be obtained directly by the application of the budget-balanced equality.

We can always write payouts as $\Pi_i = m_i(1+\rho_i)$ for some function $\rho_i$, the Return On Investment (ROI). As we require nonnegative payouts, the function $f(\cdot, m, M, \omega)$ takes values in an interval of length at most $m$, so that the ROI of any player in such a mechanism is never higher than 1 when wagers are identical.

\begin{corollary}\label{cor:roi}
If a wagering mechanism is \emph{budget-balanced}, \emph{anonymous}, \emph{truthful} and \emph{normal}, a player's ROI cannot be above 100\% when all wagers are identical.
\end{corollary}

We will see that, adding sybilproofness, this corollary generalizes to the case of different wagers.

Interestingly, when applied to the setting of Kilgour and Gerchak, our result permits to prove uniqueness of their scoring rules within the class of Competitive Prediction Schemes (CPS)~\cite{kilgour2004}. They restrict themselves to the case of eliciting the probability of a binary event; here we consider the natural generalization of their setting for the elicitation of one or more distribution properties. A CPS is essentially a wagering mechanism without a wager. In a first step each player $i$ among $n$ players makes a report $r_i$ corresponding to one or several distribution properties. In a second step, after the true outcome $\omega$ of the uncertain event becomes known, player $i$ receives a payment $\Pi_i(r_1, \dots, r_n, \omega)$ (which may be negative). The properties \emph{budget-balance}, \emph{anonymity}, \emph{truthfulness}, \emph{normality}, and \emph{individual rationality} can be directly adapted to this setting.  The
(generalized) Kilgour and Gerchak scoring rules, defined by
\begin{equation*}
\Pi_i (r_1, \dots, r_n, \omega) = s(r_i,\omega)
- \frac{1}{n-1} \sum_{j \neq i} s(r_j, \omega) ~,
\end{equation*}
where $s$ a strictly proper score function for a vector of properties $\Gamma$, can be shown to be budget-balanced, anonymous, truthful, normal, and individually rational (based on K\&G~\cite{kilgour2004} and Theorem~\ref{th:wsattributes}). Noting that the constraint of nonnegative payouts plays no role in proving Lemma~\ref{lem:tn} and~\ref{lem:tna}, Lemma~\ref{lem:tnabb} shows that these scoring rules are the \emph{only} competitive prediction schemes to satisfy these core properties.

\begin{theorem}\label{th:kg_unique}
Given $n$ players, any Competitive Prediction Scheme is \emph{budget-balanced}, \emph{anonymous}, \emph{truthful} for $\Gamma$ and \emph{normal} if and only if the payment of agent $i$ when the true outcome is $\omega$ is given by
\begin{equation*}
s(r_i,\omega) - \frac{1}{n-1} \sum_{j \neq i} s(r_j, \omega) ~,
\end{equation*}
where $s$ is strictly proper for $\Gamma$.
\end{theorem}

\subsection{Uniqueness of Weighted Score}

We now turn to the main theorem.

\begin{theorem}\label{th:charact}
A wagering mechanism is \emph{budget-balanced}, \emph{anonymous}, \emph{truthful} for $\Gamma$, \emph{normal} and \emph{sybilproof}, if and only if the payouts are given by
\[
\Pi_i(\bfr, \bfm, \omega) = m_i + m_i \left( s^M(r_i, \omega)
- \frac{1}{M} \sum_j m_j s^M(r_j, \omega) \right)
\]
where $M = \sum_j m_j$, and $s^M$ is a smooth function taking values in $[0,1]$ that is a strictly proper score function for $\Gamma$.
\end{theorem}

This result allows to complement Corollary~\ref{cor:roi}.
\begin{corollary}\label{cor:roi2}
If a wagering mechanism is \emph{budget-balanced}, \emph{anonymous}, \emph{truthful}, \emph{normal}, and \emph{sybilproof}, a player's ROI is never above 100\%.
\end{corollary}

\begin{proof}[(Theorem~\ref{th:charact})]
The proof of Theorem~\ref{th:wsattributes} may be applied directly to show that weighted score mechanisms
satisfy these properties. Here we show the other direction. The proof proceeds in three steps below.

To start, suppose that there are $n$ players wagering the same amount $m$, and let $M = n m$ be the total amount wagered. Since the wagers are identical, we know that the payout function takes the special form given in Lemma~\ref{lem:tnabb}. Let $\tilde{f}(r,m,M, \omega)$ be the function $f$ given by this lemma. Fix any possible report $r_0$ and let $f(r,m,M,\omega) = \tilde{f}(r,m,M, \omega) - \tilde{f}(r_0, m, M, \omega)$, then

\begin{eqnarray}\label{eq:charact-1}
\lefteqn{\Pi_i(\bfr, \bfm, \omega) } \nonumber
\\
&=& m + f(r_i, m, M, \omega) - \frac{1}{n-1} \sum_{j \neq i} f(r_j, m, M, \omega) ,
\end{eqnarray}
and $f(r_0,m,M,\omega) = 0$ for all $m,M,\omega$.

In the first step, we start with this equation and show that it is possible to create a function $s$ such that for any positive integers $k$ and $n$,
\begin{equation}\label{eq:charact-sfunc}
s\left( r, \omega, \frac{n}{2^k} \right)
=
2^k \frac{n}{n-1}
f \left( r, \frac{1}{2^k}, \frac{n}{2^k}, \omega \right) ~.
\end{equation}

In the second step, we show that for any number of players $n$, any set of reports $\bfr$, and any set of (not necessarily identical) wagers $\bfm$, we can write the payout to player $i$ as
\begin{equation}\label{eq:charact-payouts}
\Pi_i(\bfr, \bfm, \omega)
\!=\! m_i \left(\!\! 1\!+\! s(r_i, \omega, M)
- \frac{\sum_j m_j s(r_j, \omega, M)}{M} \!\!\right)
\end{equation}
where $M = \sum_i m_i$ and $s$ is the function defined in Step 1.

Finally, in Step 3, we show that the function $s$ can be written as a strictly proper score function taking values in $[0,1]$, completing the proof. More details on each step follow.

\ \\
\noindent\underline{\sc Step 1:}

Let $(a,b)$ be the interval of possible reports.  We begin by showing how to create a function $s: (a,b) \times \Omega \times \bbR_+ \mapsto \bbR$ that satisfies Equation~(\ref{eq:charact-sfunc}) for any positive integers $k$ and $n$.  To do so, we consider $z$ groups of $2^\ell$ players, for some $\ell > 0$. Every player in each of the groups wagers $1/(2^{k+\ell})$. In the first group, each player reports $r$, while in the other groups, each player reports $r_0$. By summing over the value of Equation~(\ref{eq:charact-1}) for each member of the first group, we see that the aggregate payout for the first group is

\begin{equation}\label{eq:charact-group1agg}
\frac{1}{2^{k}}
+ 2^\ell \left( 1 - \frac{2^\ell - 1}{z \cdot 2^\ell - 1} \right)
f\left(r, \frac{1}{2^{k+\ell}}, \frac{z}{2^k}, \omega\right) ~.
\end{equation}

Now consider an alternate situation in which a single player wagers $1/2^k$ and reports $r$ against $n-1$ other players, each wagering $1/2^k$ and reporting $r_0$. By Equation~(\ref{eq:charact-1}), the payout of the first player in this scenario is
\begin{equation}\label{eq:charact-agent1pay}
\frac{1}{2^k} + f\left(r, \frac{1}{2^k}, \frac{n}{2^k}, \omega\right)~.
\end{equation}

By sybilproofness, Equation~(\ref{eq:charact-group1agg}) must equal Equation~(\ref{eq:charact-agent1pay}) when the number of groups $z$ is $n$. Hence by simple algebra,
\begin{eqnarray}\label{eq:charact-5}
\lefteqn{2^k \left(\frac{n - 2^{-\ell}}{n-1}\right) f\left(r, \frac{1}{2^k}, \frac{n}{2^k}, \omega\right)} \nonumber
\\
&=& 2^{k+\ell} f\left(r, \frac{1}{2^{k+\ell}}, \frac{n}{2^k}, \omega\right) ~.
\end{eqnarray}

Now, since
\[\lim_{\ell \to +\infty} 2^{k} \left(\frac{n - 2^{-\ell}}{n-1}\right)
= 2^k \left(\frac{n}{n-1}\right) ~,
\]
it must be the case that the limit of the right-hand side of Equation~(\ref{eq:charact-5}) exists. Thus there exists a function $s: (a,b) \times \Omega \times \bbR_+ \mapsto \bbR$ such that
\begin{equation*}
s\left(r, \omega, \frac{n}{2^k}\right)
= \lim_{\ell \to +\infty} 2^{k+\ell} f\left(r, 2^{-(k+\ell)}, \frac{n}{2^k}, \omega\right) ~.
\end{equation*}
Plugging this into Equation~(\ref{eq:charact-5}) yields Equation~(\ref{eq:charact-sfunc}).

\ \\
\noindent\underline{\sc Step 2:}

Using the previous result, we now show that for any number of players $n$, for any reports and (not necessarily identical) wagers, the payout to each player $i$ can be expressed as in Equation~(\ref{eq:charact-payouts}).  To do so we use a continuity argument applied to an approximation of the payouts. In particular, we assume that the vector of wagers belongs to the set
\[
\scM = \{ (a_1 2^{-k}, \dots, a_n 2^{-k}) / \bfa \in \bbN^n, k \in \bbN \} ~.
\]
While this assumption does restrict the set of possible wagers, we note that because $\scM$ is dense in $\bbR_+^n$, it is the case that for \emph{any} positive vector of wagers $\bfm$, there exist elements of $\scM$ arbitrarily close to $\bfm$.  Thus any vector of wagers can be approximated arbitrarily well using a vector in $\scM$.

Now, given a vector of wagers $\bfm \in \scM$, there clearly exist a vector of integers $\bfa$ and a value $k > 0$ such that $m_i = a_i/2^k$ for all $i$. Let $N = \sum_i a_i$. To obtain the vector of payouts $\Pi(\bfr, \bfm, \omega)$, we compare the present scenario with another in which there are $N$ players divided into $n$ groups. Each group $i$ contains $a_i$ players, each wagering $1/2^k$ on report $r_i$. By Equation~(\ref{eq:charact-1}) and~(\ref{eq:charact-sfunc}) the aggregate payout of group $i$ in this alternate scenario is
\begin{eqnarray*}
\lefteqn{\frac{a_i}{2^k}
+  a_i \Bigg( \left( 1 - \frac{a_i - 1}{N-1} \right) f\left(r_i, \frac{1}{2^k}, \frac{N}{2^k}, \omega\right) }
\\
& & \qquad - \frac{1}{N-1} \sum_{j \neq i} a_j f\left(r_j, \frac{1}{2^k}, \frac{N}{2^k}, \omega\right) \Bigg)
\\
&=& \frac{a_i}{2^k} + a_i \Bigg(\frac{N-a_i}{2^k N} s(r_i, \omega, M)
\\
& &\qquad \qquad \quad - \sum_{j \neq i} \frac{a_j}{2^k N} s(r_j, \omega, M) \Bigg)
\\
&=& \frac{a_i}{2^k} \left(1 + s(r_i, \omega, M) - \sum_j \frac{a_j}{N} s(r_j, \omega, M) \right)
\\
&=& m_i \left(1 + s(r_i, \omega, M) - \frac{1}{M} \sum_j m_j s(r_j, \omega, M) \right) .
\end{eqnarray*}

By sybilproofness, this quantity is precisely $\Pi_i(\bfr, \bfm, \omega)$, the payout of player $i$ in the initial scenario with $n$ players.  Thus since the set $\scM$ is dense in $\bbR_+^n$, by continuity of the payout functions, for all $\bfm \in \bbR_+^n$, Equation~(\ref{eq:charact-payouts}) holds.

\ \\
\noindent\underline{\sc Step 3:}

It remains to show that the payout can always be written as in Equation~(\ref{eq:charact-payouts}) using a function $s$ that is a score function taking values in $[0,1]$.  First, notice that $s(\cdot,\omega, M)$ must be bounded, since payouts are always nonnegative. Let $L(\omega, M) = \inf s(\cdot,\omega, M)$, and let $\tilde{s}(r, \omega, M) = s(r,\omega, M) - L(\omega, M)$. Note that we can write
\[
\Pi_i(\bfr, \bfm, \omega) = m_i + m_i \left( \tilde{s}(r_i, \omega, M) - \frac{\sum_j m_j \tilde{s}(r_j, \omega, M)}{M} \right) ~.
\]

Consider the scenario in which there are only two players, one wagering $\epsilon$ and reporting $r$, and the other wagering $M - \epsilon$ and reporting $r'$. The payout of the first player is
\[
\epsilon \left[ 1 + \tilde{s}(r, \omega, M) - \frac{\epsilon \tilde{s}(r, \omega, M) + (M-\epsilon) \tilde{s}(r', \omega, M)}{M} \right] \geq 0~.
\]
Since $\inf \tilde{s}(\cdot, \omega, M) = 0$, for all $r'$,
\[
1 - \frac{(M-\epsilon) \tilde{s}(r', \omega, M)}{M} \geq 0~,
\]
and, by taking the limit as $\epsilon \to 0$, for all $r'$,
\[
\tilde{s}(r', \omega, M) \leq 1~.
\]
Therefore $\tilde{s}$ takes values in $[0,1]$.  The requirement of truthfulness implies that, for all $M > 0$, $\tilde{s}(\cdot, \cdot, M)$ is strictly proper for $\Gamma$, which concludes the proof.
\end{proof}

\section{Extensions}
\label{sec:discussion}

We now present several extensions of the weighted-score mechanisms, each of which achieves properties that weighted-score does not at the expense of other properties.

\subsection{Adaptive weighted-score}
\label{subsec:adaptive}

One major difficulty encountered in designing score functions is that of incentive calibration: scores should vary the most in regions that are more likely to contain the true property value~\cite{savage1971,winkler1996}. As an example, consider designing a score function to collect reports about expected points in sports games. Without precise knowledge of the teams, the game, and its statistics, one must consider a large interval of possible point values. Yet informed forecasters are likely to report points within a small window. This mismatch induces small reward differences amongst forecasters. It also reduces incentives for agents to reveal their own belief with precision. However given expert advice limiting the plausible property values in advance, a score function with a large reward range for forecasters can be created.

The problem of properly adjusting score functions is particularly important for the weighted-score mechanism. Indeed, if score differences are small, the amplitude of money transfers between participants is likely to be considerably lower than initial wagers, thereby reducing incentives to participate.

We propose of a variant of weighted-score mechanism with self-adjusting score functions. To do this, consider a decomposition of the initial game into multiple, smaller games where only half of the players participate with a small fraction of their wagers. For each of these games, the payouts are computed according to a weighted-score payout function whose score functions are parameterized by sets of property values. This set, sent from the other half of the players, provides very accurate information about the regions where most reports are made. In any of the smaller games, players who participate do not influence the shape of the payout function, so that truthfulness remains true, although sybilproofness is lost.

We now give the formal definition. Given a set of $n$ players $\scN$, we consider the set of groups of players of size $\lceil n/2 \rceil$, $\scS = \{ A \subseteq \scN / |A| = \lceil n/2 \rceil\}$, and $\scS_i$ the set of sets of $\scS$ containing $i$. For any set of property values $R$ (provided, for examples, by experts as likely property values), let $s^R$ be a strictly proper score function for $\Gamma$. Let $m_i' = m_i/|\scS_i|$ be the wager of player $i$ in each ``small game''. We call \emph{adaptive} weighted-score mechanism a wagering mechanism whose payouts are
\begin{align*}
\Pi_i(\bfr, \bfm, \omega) = m_i + & \sum_{S \in \scS_i} m_i' \Bigg[ s^{\{r_\ell/\ell \in \scN\backslash S\}}(r_i, \omega) \\
& - \frac{\sum_{j \in S} m_j' s^{\{r_\ell/\ell \in \scN\backslash S\}}(r_j, \omega) }{\sum_{j \in S} m_j'} \Bigg]
\end{align*}

Compared to the original weighted-score mechanism, this variant loses normality and sybilproofness, but maintains the other properties.\footnoteremember{mytheonote}{Theorem~\ref{th:adaptive}, \ref{th:rando} and~\ref{th:pmscore} may be proved in a similar fashion as Theorem~\ref{th:wsattributes}. We omit the proofs due to space restrictions.}
\begin{theorem}\label{th:adaptive}
Adaptive weighted-score mechanisms are \emph{budget-balanced}, \emph{anonymous}, \emph{truthful}, \emph{individually rational} and \emph{monotonic}.
\end{theorem}

\subsection{Higher stakes}

The notion of ``betting $m$ dollars'' is commonly interpreted as placing a bet such that, in the worst case, $m$ dollars will be lost. In the weighted-score mechanisms discussed here, players may not be able to lose their entire wager, no matter what the other players do.  Indeed, when a player reports a value that does not minimize the score function, no matter what the outcome, they are guaranteed to recoup part of their wager.  An important case is that of eliciting the probability of a binary event: Theorem~\ref{th:charact} demonstrates that these ``low stakes'' are true for any mechanism having the core properties; any player announcing a probability other than 0 or 1 is certain to recoup part of her wager.  Randomization can address this drawback.

Let $\Gamma=(\Gamma_1, \dots, \Gamma_k)$ be a vector of distribution properties, $s$ be strictly proper for $\Gamma$, and $\tilde{s}$ be proper (but not necessarily strictly proper), that is,
\begin{equation*}
E_P[\tilde{s}(r, \omega)] \leq E_P[\tilde{s}(\Gamma(P), \omega)]
\end{equation*}
for all admissible reports $r$ and probabilities $P$.

Consider computing the payout function as follows. First, flip a coin. If the coin is heads, then for all $i$,
\[
\Pi_i(\bfr, \bfm, \omega) = m_i + m_i \left( s(r_i, \omega) - \frac{\sum_j s(r_j, \omega) m_j}{\sum_j m_j} \right) ~.
\]
Otherwise, for all $i$,
\[
\Pi_i(\bfr, \bfm, \omega) = m_i + m_i \left( \tilde{s}(r_i, \omega) - \frac{\sum_j \tilde{s}(r_j, \omega) m_j}{\sum_j m_j} \right) .
\]

Randomization is equivalent to inserting a factor in the outcome space, using $\Omega' = \Omega \times \{ H, T \}$, where $\{H, T\}$ is the outcome space for the coin flip. As long as the probabilities of heads and tails are strictly positive, the randomized mechanism conserves all of the properties of weighted-score wagering.\footnoterecall{mytheonote}

\begin{theorem}\label{th:rando}
The randomized weighted-score mechanism is \emph{budget-balanced}, \emph{anonymous}, \emph{truthful}, \emph{individually rational}, \emph{sybilproof}, \emph{normal} and \emph{monotonic}.
\end{theorem}

If the alternative score only takes extreme values, a player can always lose her wager.\footnote{At the limit as the total wager of other players grows.} In the case of eliciting the probability of an event, one may choose
\begin{equation*}
\tilde{s}(p,\omega) = \begin{cases}
1 & \text{ if } |p-\omega| \leq \frac{1}{2}~, \\
0 & \text{ otherwise.}
\end{cases}
\end{equation*}

\subsection{Higher ROI}

By Corollaries~\ref{cor:roi} and~\ref{cor:roi2}, no wagering mechanism satisfying the core properties can result in returns on investment higher that 100\%. This contrasts with most betting markets, wherein players may win several times their wager by betting on a very unlikely outcome. To solve this, one may use parimutuel-score mechanisms, where payouts are computed according to
\begin{equation*}
\Pi_i(\bfr, \bfm, \omega) = \frac{m_i s(r_i, \omega)}{\sum_j m_j
s(r_j, \omega)}\sum_j m_j
\end{equation*}
where $s$ is a strictly proper score function for $\Gamma$.

Parimutuel-score mechanisms are very similar to certain mechanisms introduced by Johnstone~\cite{johnstone2007} and may be used in the same context. Like horse-race type parimutuel betting markets, parimutuel-score schemes have unbounded return on investment. They conserve several desirable properties.\footnoterecall{mytheonote}
\begin{theorem}\label{th:pmscore}
Parimutuel-score mechanisms are \emph{budget-balanced}, \emph{anonymous}, \emph{sybilproof} and \emph{monotonic}.
\end{theorem}
However, they are \emph{not truthful}, \emph{not individually rational} and \emph{not normal}. They may be considered as approximately truthful in many practical situations: indeed, asymptotically as the number of players grows, a player tends to lose influence over the denominator of the payout function. The payout then becomes proportional to the strictly proper score function $s$, which ensures truthfulness.

\section{Conclusion}

We have investigated wagering mechanisms for revealing individual predictions from a group of agents. Agents are called to report on some information about some random experiment. Along with their report, they place a wager in a common pot. Upon realization of the experiment, agents receive a payment that depends on the true outcome and their own report. Payments are composed of a share of the common pot and possibly bonuses or taxes/fees. These mechanisms include many instances of common betting markets and call markets.

We have identified a particular subclass of such mechanisms called weighted-score mechanisms. Those novel elicitation schemes provide free individual forecasts of distribution properties, such as probabilities of binary events, or expectations and quantiles of random variables. They satisfy a number of desirable properties, including budget-balance, anonymity, truthfulness, normality, sybilproofness, individual rationality, and monotonicity.

Furthermore, we have showed that weighted-score mechanisms, parameterized by the total money wagered, are the only mechanisms that are budget-balanced, anonymous, truthful, normal, and sybilproof. In addition, we have proved that the Kilgour and Gerchak's scoring rules are the only forecasting methods to satisfy the core properties (for the setting considered in~\cite{kilgour2004}).

We have explored variants of weighted-score mechanisms that conserve many desirable properties, and improve the core mechanism in several ways, by offering adaptive rewards, higher stakes, and unbounded returns on investment.

Our theoretical investigation leaves several open paths for future research. An important question, not addressed in the present work, is that of empirical studies. How do our mechanisms perform in a realistic environment? How do people behave when faced with a weighted-score mechanism or one of its variants? Also, our paper concentrated on one-shot mechanisms. A natural important step is to introduce dynamism: can we develop similar mechanisms -- and similar characterizations -- when adding a time dimension? In particular, is it possible to conserve truthfulness (and so avoid bluffing strategies), and incentivize agents to reveal their prediction early? This would permit to elicit at all times individual beliefs of agents, and watch their aggregation.

\section*{Acknowledgments}

The authors would like to thank Esteban Arcaute, Alina Beygelzimer, Preston McAfee, Bethany Soule and Yevgeniy Vorobeychik for helpful discussions and suggestions, and the anonymous reviewers for their useful comments.

\newpage
\bibliographystyle{abbrv}


\end{document}

